Author’s response to reviews

Title: Design of the Growth hormone deficiency and Efficacy of Treatment (GET) score and non-interventional proof of concept study

Authors:

Peter Kann (kannp@med.uni-marburg.de)
Simona Bergmann (simona.bergmann@med.uni-marburg.de)
Martin Bidlingmaier (Martin.Bidlingmaier@med.uni-muenchen.de)
Christina Dimopoulou (chdimo@hotmail.com)
Birgitte Pedersen (BITP@novonordisk.com)
Günter Stalla (stalla@mpipsykl.mpg.de)
Matthias Weber (mmweber@uni-mainz.de)
Stefanie Meckes-Ferber (STMF@novonordisk.com)

Version: 2 Date: 15 Jan 2018

Author’s response to reviews:

Dear Reviewers,

Thank you for the additional review comments. Responses to each of the comments follow:

1) We confirm that all participants were aged over 18 years.

2) Individual author names have been represented using initials (page 17, line 408)

3) The "Additional files" section has been moved after the References/Figure legends (page 21, line 515).

4) A clean version of the manuscript has been uploaded.